

A peridomestic Aedes malayensis population in Singapore can transmit yellow fever virus

Elliott F Miot, Fabien Aubry, Stéphanie Dabo, Ian H Mendenhall, Sébastien Marcombe, Cheong H Tan, Lee C Ng, Anna-Bella Failloux, Julien Pompon, Paul T Brey, et al.

▶ To cite this version:

Elliott F Miot, Fabien Aubry, Stéphanie Dabo, Ian H Mendenhall, Sébastien Marcombe, et al.. A peridomestic Aedes malayensis population in Singapore can transmit yellow fever virus. PLoS Neglected Tropical Diseases, 2019, 13 (10), pp.e0007783. 10.1371/journal.pntd.0007783. pasteur-02325668

HAL Id: pasteur-02325668 https://pasteur.hal.science/pasteur-02325668

Submitted on 22 Oct 2019

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



GOPEN ACCESS

Citation: Miot EF, Aubry F, Dabo S, Mendenhall IH, Marcombe S, Tan CH, et al. (2019) A peridomestic *Aedes malayensis* population in Singapore can transmit yellow fever virus. PLoS Negl Trop Dis 13 (10): e0007783. https://doi.org/10.1371/journal. pntd.0007783

Editor: Elvina Viennet, Australian Red Cross Blood Service, AUSTRALIA

Received: July 10, 2019

Accepted: September 15, 2019

Published: October 7, 2019

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0007783

Copyright: © 2019 Miot et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

RESEARCH ARTICLE

A peridomestic *Aedes malayensis* population in Singapore can transmit yellow fever virus

Elliott F. Miot^{1,2,3}*, Fabien Aubry¹, Stéphanie Dabo¹, Ian H. Mendenhall⁴, Sébastien Marcombe³, Cheong H. Tan⁵, Lee C. Ng⁵, Anna-Bella Failloux⁶, Julien Pompon^{4,7}, Paul T. Brey^{3©}, Louis Lambrechts^{1©}*

1 Insect-Virus Interactions Unit, Institut Pasteur, UMR2000, CNRS, Paris, France, 2 Sorbonne Université, Collège doctoral, Paris, France, 3 Medical Entomology and Vector-Borne Disease Unit, Institut Pasteur du Laos, Vientiane, Lao PDR, 4 Program in Emerging Infectious Disease, Duke-NUS Medical School, Singapore, 5 Environmental Health Institute, National Environment Agency, Singapore, 6 Arboviruses and Insect Vectors Unit, Institut Pasteur, Paris, France, 7 MIVEGEC, IRD, CNRS, Univ. Montpellier, Montpellier, France

• These authors contributed equally to this work.

* elliott.miot@gmail.com (EFM); louis.lambrechts@pasteur.fr (LL)

Abstract

The case-fatality rate of yellow fever virus (YFV) is one of the highest among arthropodborne viruses (arboviruses). Although historically, the Asia-Pacific region has remained free of YFV, the risk of introduction has never been higher due to the increasing influx of people from endemic regions and the recent outbreaks in Africa and South America. Singapore is a global hub for trade and tourism and therefore at high risk for YFV introduction. Effective control of the main domestic mosquito vector Aedes aegypti in Singapore has failed to prevent re-emergence of dengue, chikungunya and Zika viruses in the last two decades, raising suspicions that peridomestic mosquito species untargeted by domestic vector control measures may contribute to arbovirus transmission. Here, we provide empirical evidence that the peridomestic mosquito Aedes malayensis found in Singapore can transmit YFV. Our laboratory mosquito colony recently derived from wild Ae. malayensis in Singapore was experimentally competent for YFV to a similar level as Ae. aegypti controls. In addition, we captured Ae. malayensis females in one human-baited trap during three days of collection, providing preliminary evidence that host-vector contact may occur in field conditions. Finally, we detected Ae. malayensis eggs in traps deployed in high-rise building areas of Singapore. We conclude that Ae. malayensis is a competent vector of YFV and re-emphasize that vector control methods should be extended to target peridomestic vector species.

Author summary

Yellow fever is a dreadful disease caused by a mosquito-borne virus circulating in Africa and South America. Historically, the Asia-Pacific region has remained free of yellow fever but the ever increasing influx of travelers puts places such as Singapore at unprecedented risk of yellow fever virus introduction. The present study characterized the potential contribution of a mosquito species called *Aedes malayensis* to yellow fever virus transmission Funding: EFM was supported by a Calmette and Yersin doctoral fellowship of the Institut Pasteur International Network. This work was funded by Agence Nationale de la Recherche (grant ANR-17-ERC2-0016-01 to LL), the French Government's Investissement d'Avenir program Laboratoire d'Excellence Integrative Biology of Emerging Infectious Diseases (grant ANR-10-LABX-62-IBEID to LL), the European Union's Horizon 2020 research and innovation program under ZikaPLAN grant agreement no. 734584 (to LL), and the City of Paris Emergence(s) program in Biomedical Research (to LL). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

in Singapore. *Aedes malayensis* breeds in urban parks of Singapore and is suspected to have participated in the resurgence of other mosquito-borne diseases such as dengue because it is not targeted by current mosquito control measures. Not only was *Ae. malayensis* able to experimentally acquire and transmit yellow fever virus, but it was also found to engage in contact with humans in a field situation. This empirical evidence indicates that *Ae. malayensis* is a competent vector of yellow fever virus and should be targeted by mosquito control programs.

Introduction

The case-fatality rate of yellow fever virus (YFV) ranges from 15% to 50% [1] and is one of the highest among arthropod-borne viruses (arboviruses). YFV is endemic in 47 countries in Africa and South America, with an estimated annual incidence of around 200,000 cases and 30,000 deaths [2]. Despite an efficient vaccine against YFV, the past few years have seen a growing number of YFV outbreaks (Democratic Republic of the Congo, Angola, Uganda, Brazil and most recently Nigeria) [3–5]. During such outbreaks, increasing numbers of unvaccinated travelers who become infected and return to non-endemic countries have raised the risk of YFV introduction to unprecedented levels [6]. The Asia-Pacific region has remained free of YFV until now but the risk of introduction has never been higher [6–8]. In 2016, eleven Chinese workers infected with YFV in Angola who returned to China were the first cases of YFV diagnosed in travelers to Asia [7]. Over two billion immunologically naïve people live in Asia and the current vaccine production capacity would be insufficient to prevent a massive YFV epidemic, while mosquito control programs would be overwhelmed [8].

With 18.5 million visitors in 2018, Singapore is a global hub for tourism, trade and transport that includes one of the busiest ports in the world and a major international airport. These features put Singapore at high risk for introduction of arboviruses, including YFV. Strict and sustained vector control measures together with household structural improvements, currently achieve very low densities of the domestic arbovirus vector *Aedes aegypti* [9]. Re-emergence of dengue virus (DENV), chikungunya virus (CHIKV) and Zika virus in Singapore in the last two decades [10–13] supports the hypothesis that peridomestic mosquito species untargeted by vector control measures may contribute to 'cryptic' arbovirus transmission [14]. In particular, the peridomestic mosquito *Aedes malayensis* (a member of the *Stegomyia* subgenus) breeds in urban parks of Singapore and is experimentally competent for DENV and CHIKV [14].

Here, we evaluated the potential contribution of the peridomestic mosquito *Ae. malayensis* to YFV transmission in Singapore. Vector competence assays in the laboratory and a small-scale field survey provided evidence that indeed *Ae. malayensis* could contribute to YFV transmission in Singapore.

Methods

Ethics statement

This study used human blood samples to prepare mosquito artificial infectious blood meals. Healthy donor recruitment was organized by the local investigator assessment using medical history, laboratory results and clinical examinations. Biological samples were supplied through participation of healthy volunteers at the ICAReB biobanking platform (BB-0033-00062/ ICAReB platform/Institut Pasteur, Paris/BBMRI AO203/[BIORESOURCE]) of the Institut Pasteur to the CoSImmGen and Diagmicoll protocols, which have been approved by the French Ethical Committee Ile-de-France I. The Diagmicoll protocol was declared to the French Research Ministry under reference DC 2008–68 COL 1. The use of human-baited double net trap was approved by the National Environment Agency (NEA) of Singapore (NEA/ PH/CLB/19-00004). All adult subjects provided written informed consent.

Mosquitoes

Experiments were carried out with a laboratory colony derived in 2014 from a wild population of *Ae. malayensis* in Singapore and subsequently maintained at the Duke-NUS Medical School for >50 generations [14]. The 8th generation of a laboratory colony of *Ae. aegypti* maintained at the Institut Pasteur in Laos was used as a control. The *Ae. malayensis* colony was initiated with mosquito eggs collected from two forested areas located in the Northern (Sembawang) and the Southern (East Coast Park) regions of Singapore [14]. The *Ae. aegypti* colony originated in the town of Paksan, Paksan district, Bolikhamsai province, Laos. Mosquitoes were reared under controlled insectary conditions (28°C, 70% relative humidity, 12:12 hour light cycle). Eggs were hatched synchronously in a vacuum chamber for 1 hour. Larvae were reared in 24 x 34 x 9 cm plastic trays containing 1.5 L of dechlorinated tap water and supplemented with Tetramin (Tetra) fish food at a density of 400 larvae per tray. Six hundred adults were kept in 30 x 30 x 30 cm Bugdorm-1 insect cages with permanent access to 10% sucrose solution.

Virus

Vector competence experiments were carried out with a low-passage YFV isolate (YFV-S79 strain) belonging to the West African lineage, which was originally obtained in 1979 from the serum of a patient returning to France from Senegal [15]. Prior to its use in the experiments, the YFV isolate was passaged twice in newborn mouse brains and three times in *Aedes albopic-tus* C6/36 cells. Virus stock was produced during the last passage, as previously described for DENV [16]. Virus titration was performed by standard focus-forming assay (FFA), as previously described for DENV [16] with the exception of the primary antibody. A mouse anti-flavivirus group antigen monoclonal antibody MAB10216 (Merck Millipore) diluted 1:1,000 in phosphate-buffered saline supplemented with 1% bovine serum albumin (Interchim) was used as primary antibody.

Virus challenge

In two separate experiments (referred to as experiments 1 and 2 hereafter), mosquitoes were orally challenged with YFV as previously described for DENV [16]. Briefly, eight-day-old females deprived of sucrose for 24 hours were offered an artificial infectious blood meal in 3 rounds of 15 min using a Hemotek membrane feeding apparatus with porcine intestine as the membrane. Blood meals consisted of a 2:1 mix of washed human erythrocytes and virus suspension. Adenosine triphosphate (Merck) was added as a phagostimulant [17] to the blood meal at a final concentration of 10 mM. To obtain a dose-response curve, three separate blood meals with increasing YFV concentrations were prepared. In experiment 1, mosquitoes were exposed to 2.9 x 10⁶ focus-forming units (FFUs)/mL, 2.1 x 10⁵ FFUs/mL, or 1.8 x 10⁴ FFUs/mL or 3.6 x 10³ FFUs/mL of YFV. Fully engorged females were sorted on wet ice, transferred into 0.5-L cardboard containers and maintained in a climatic chamber (28°C, 70% relative humid-ity, 12:12 hour light cycle) with permanent access to 10% sucrose solution. At 10 and 14 days post blood meal, mosquitoes were cold anesthetized (experiment 1) or paralyzed with

triethylamine (experiment 2) to perform *in vitro* salivation [18]. In experiment 1, the wings and legs were removed and stored at -80 °C prior to the salivation assay. In experiment 2, the heads were removed and stored at -80 °C after the salivation assay. The proboscis of each female was inserted into a 20- μ L pipet tip containing 5 μ L of fetal bovine serum (FBS). After 30 min of salivation, the body remainders were stored dry at -80 °C. The saliva-containing FBS was mixed with 45 μ L of Leibovitz's L-15 medium and immediately inoculated onto C6/36 cells for titration by FFA, as described above, without subsequent dilution.

Virus detection

Bodies were homogenized individually in 300 μ L of lysis buffer (Tris 10 mM, NaCl 50 mM, EDTA 1.27 mM with a final pH adjusted to 8.2) supplemented with proteinase K (1 μ L for 55.5 μ L of buffer). Body homogenates were centrifuged and 100 μ L of supernatant was incubated for 5 min at 56°C then 10 min at 98°C to extract viral RNA. Detection of YFV RNA was performed using a two-step RT-PCR reaction to generate a 192-bp amplicon located in a conserved region of the *NS3* gene of YFV. Total RNA was reverse transcribed into cDNA with random hexamers using M-MLV reverse transcriptase (ThermoFisher Scientific) using the following program: 10 min at 25°C, 50 min at 37°C and 15 min at 70°C. The cDNA was subsequently amplified using DreamTaq DNA polymerase (ThermoFisher Scientific). The 20- μ L reaction volume contained 1x of reaction mix and 10 μ M of primers (forward: 5'-GCGTAAG GCTGGAAAGAGTG-3'; reverse: 5'-CTTCCTCCCTTCATCCACAA-3'; [19]). The thermocycling program was 2 min at 95°C, 35 cycles of 30 sec at 95°C, 30 sec at 60°C, and 30 sec at 72°C with a final extension step of 7 min at 72°C. Amplicons were visualized by electrophoresis on a 2% agarose gel. Mosquito wings/legs and heads were homogenized in 300 μ L of Leibovitz's L-15 medium supplemented with 2% FBS, centrifuged, and titrated by FFA as described above.

Host-vector contact

Adult mosquito sampling was conducted in a forested area in the northern part of Singapore (Sembawang; 1.456493, 103.825983) in March 2019. Host-vector contact was evaluated using a human-baited double net trap as previously described [20]. Briefly, a human volunteer was resting on a long chair inside a small, untreated bed-net protected from mosquitoes during daylight hours (7 a.m. to 7 p.m.). A larger bed-net covered the smaller one with a 30-cm opening at the bottom allowing mosquitoes to enter the 20-cm space between bed-nets. A large plastic sheet was placed above the bed-nets as a protection. Every 10 min, the human volunteer collected mosquitoes trapped between bed-nets using a mouth aspirator. Aspirated mosquitoes were transferred into holding cups, killed and grouped per 1-hour interval. Back in the laboratory, mosquitoes were morphologically identified and stored at -80°C.

Gravitrap survey

The presence of *Ae. malayensis* in Singapore's high-rise public housing was assessed using surveillance data obtained in 2018 from the Singapore NEA. The island-wide Gravitrap surveillance network was implemented by NEA to monitor the spatial and temporal variability of adult *Aedes* mosquito populations and to evaluate the effectiveness of vector control measures. In 2018, approximately 53,000 Gravitraps were deployed in about 9,200 public apartment blocks across the island. The Gravitrap is a simple, hay infusion-filled cylindrical trap with a sticky inner surface to catch ovipositing gravid female *Aedes* mosquitoes [21].

Statistical analyses

Vector competence was estimated by measuring virus infection, dissemination and transmission rates. Infection rate was estimated as the proportion of YFV-positive mosquitoes among all mosquitoes tested. Dissemination rate was estimated as the proportion of infected mosquitoes with YFV-positive wings and legs (experiment 1) or head (experiment 2). Transmission rate was estimated as the proportion of mosquitoes with a disseminated infection that had YFV-positive saliva. In addition, transmission efficiency was calculated as the proportion of mosquitoes with infectious saliva among all mosquitoes tested. Vector competence indices were analyzed with a logistic regression model in which each individual mosquito was associated with a binary variable (1 = YFV-positive or 0 = YFV-negative), followed by an analysis of deviance with the R package *car* [22]. An initial analysis showed that infection rates were not statistically significantly different between 10 and 14 days post blood meal and both time points were subsequently combined. The analysis of dissemination and transmission rates was restricted to 14 days post blood meal because the two time points differed and there was too little data on day 10 for a meaningful analysis. The experiment effect was statistically non-significant overall and removed from the model. For infection and dissemination rates, the final regression model included the infectious dose (\log_{10} -transformed blood meal titer), the mosquito species (Ae. aegypti and Ae. malayensis), and their interaction. The 50% oral infectious dose (OID₅₀) and 10% oral infectious dose (OID₁₀) values and their respective 95% confidence intervals were derived from the logistic fits of infection rates using the R package MASS [23]. The analysis of transmission rate was restricted to the highest infectious dose and the final regression model only included the mosquito species. All statistical analyses were performed using the R software version 3.5.2 [24] and graphical representations were generated with the R packages ggplot2 [25] and ggpubr [26].

Results

We measured the vector competence of 68 and 31 *Ae. malayensis* females and 132 and 76 *Ae. aegypti* females in experiments 1 and 2, respectively. The data from both experiments were combined because initial analyses showed that none of the vector competence indices differed significantly between them. In each experiment, mosquitoes were exposed to three different YFV infectious doses to establish a dose-response curve. Vector competence was analyzed 14 days post blood meal, with the exception of infection rates that also included mosquitoes sampled at 10 days post blood meal.

The proportion of blood-fed females that became infected (i.e., the infection rate) increased with the infectious dose (Fig 1). The highest infectious dose (2.9 x 10⁶ FFUs/mL) resulted in 74.5% and 75.6% infected females for *Ae. malayensis* and *Ae. aegypti*, respectively. Infectious dose was the only statistically significant (p < 0.0001) predictor of infection rate (Table 1), although there was a slight difference in the shape of the dose-response curve between the two species (Fig 1). Although the OID₅₀ was similar between the two species, the OID₁₀ was lower for *Ae. aegypti*. The OID₅₀ was 5.97 log₁₀ FFUs/mL (95% CI = 5.81–6.14 log₁₀ FFUs/mL) for *Ae. aegypti* and 6.13 log₁₀ FFUs/mL (95% CI = 6.02–6.25 log₁₀ FFUs/mL) for *Ae. malayensis*. The OID₁₀ was 5.03 log₁₀ FFUs/mL (95% CI = 4.78–5.28 log₁₀ FFUs/mL) for *Ae. aegypti* and 5.59 log₁₀ FFUs/mL (95% CI = 5.41–5.78 log₁₀ FFUs/mL) for *Ae. malayensis*.

The proportion of infected females that developed a disseminated infection (i.e., the dissemination rate) also increased with infectious dose (Fig 1). The highest infectious dose resulted in 42.9% and 57.1% dissemination in *Ae. malayensis* and *Ae. aegypti*, respectively. Infectious dose was the only statistically significant (p < 0.0001) predictor of dissemination



Fig 1. Peridomestic *Ae. malayensis* in Singapore are orally susceptible to YFV. Dose-response curves are shown for the *Ae. malayensis* population from Singapore under study and a control *Ae. aegypti* population. Infection rate is the proportion of blood-fed females testing YFV-positive 10–14 days post blood meal. Dissemination rate is the proportion of YFV-infected females with YFV-positive wings/legs or head 14 days post blood meal. The line represents the logistic regression of the data and the shaded area is the 95% confidence interval of the fit. The data shown are combined from two separate experiments.

https://doi.org/10.1371/journal.pntd.0007783.g001

rate (Table 1). Together, these results indicated that although *Ae. aegypti* was slightly more susceptible than *Ae. malayensis* at low infectious doses, the two species had a similar YFV susceptibility overall.

The proportion of females with a disseminated infection that had YFV-positive saliva (i.e., the transmission rate) was measured at 14 days post blood meal only for the highest infectious dose (Fig 2). YFV was detected in the saliva of 33.3% (3/9) of *Ae. malayensis* females and 37.5% (3/8) of *Ae. aegypti* females, and the difference between the two species was not statistically significant (p = 0.8576). Vector competence can be summarized by the overall proportion of blood-fed females that had YFV-positive saliva at 14 days post blood meal at the highest infectious dose (i.e., the transmission efficiency). Transmission efficiency was 11.5% (3/26) for *Ae. malayensis* and 14.3% (3/21) for *Ae. aegypti* (Fig 2), and the difference between the two species was not statistically significant (p = 0.7795).

Table 1. Test statistics of YFV infection and dissemination rates.	Infection and dissemination rates were analyzed by logistic regression. The	model included the effect
of the YFV oral infectious dose (log10-transformed blood meal titer)	, the mosquito species (Ae. aegypti or Ae. malayensis) and their interaction.	

	Infection rate			Dissemination rate		
	$LR \chi^2$	Df	p value	$LR \chi^2$	Df	<i>p</i> value
Dose	97.42	1	< 0.0001	5.504	1	0.0190
Species	3.325	1	0.0682	1.040	1	0.3079
Dose x Species	3.099	1	0.0783	1.001	1	0.3171

Df: degrees of freedom; LR: likelihood ratio.

https://doi.org/10.1371/journal.pntd.0007783.t001

Fig 2. Peridomestic *Ae. malayensis* **in Singapore are competent for YFV transmission**. Transmission potential is shown for the *Ae. malayensis* population from Singapore under study and a control *Ae. aegypti* population 14 days after oral exposure to 2.9 x 10⁶ FFU/mL of YFV. Transmission rate is the proportion of females with a disseminated infection that had infectious YFV in their saliva 14 days post blood meal. Transmission efficiency is the proportion of all blood-fed females that had infectious YFV in their saliva 14 days represent 95% confidence intervals of the proportions. The data shown are from two separate experiments.

https://doi.org/10.1371/journal.pntd.0007783.g002

In addition to vector competence, the ability of *Ae. malayensis* to transmit YFV depends on its propensity to bite humans. To assess host-vector contact of peridomestic *Ae. malayensis* in Singapore, we surveyed day-biting mosquitoes in a forested area of Sembawang, Singapore in

Fig 3. Peridomestic *Ae. malayensis* in Singapore are attracted to humans. (A) Human-baited double net trap setup in the forested area of Sembawang, Singapore. (B) Distribution of mosquito species among 115 female specimens collected at daytime in the human-baited double net trap.

https://doi.org/10.1371/journal.pntd.0007783.g003

March 2019 using a human-baited double net trap (Fig 3A). A total of 115 female mosquitoes were captured during three days of collection. The most abundant species was *Ae. albopictus* (89%) followed by *Ae. malayensis* (9%) (Fig 3B).

Out of 504,771 mosquitoes collected in Gravitraps deployed by NEA in Singapore's highrise public apartment blocks in 2018, the predominant mosquito species was *Ae. aegypti* (80% of all captures). However, a total of 1,931 *Ae. malayensis* (0.38%) were caught in 1,741 traps. The average capture rate was 0.0013 *Ae. malayensis* per trap and per week, with an average of 1.1 specimen per *Ae. malayensis*-positive trap.

Discussion

Our study indicates that peridomestic *Ae. malayensis* mosquitoes can contribute to YFV transmission in Singapore based on two lines of evidence. First, our laboratory mosquito colony recently derived from a wild *Ae. malayensis* population in Singapore was experimentally competent for YFV, to a level that was not statistically different from that of the *Ae. aegypti* controls. Adding to earlier data demonstrating vector competence of this colony for DENV and CHIKV [14], our results are the first experimental evidence, to our knowledge, that peridomestic *Ae. malayensis* from Singapore can potentially transmit YFV acquired from an infectious blood meal. Second, we captured *Ae. malayensis* females in a human-baited trap. Although our hostseeking experiment lacked a non-baited control trap, it provided preliminary evidence that this peridomestic *Ae. malayensis* population in Singapore may engage in human-biting behavior, as was already observed for other populations of the species in South-East Asia [27,28].

Despite its wide distribution in South-East Asia with records in Laos, Thailand, Cambodia, Vietnam, Peninsular Malaysia, the Andaman and Nicobar Islands, and Taiwan [27-30], the mosquito Ae. malayensis has been relatively poorly studied. Recently, however, Ae. malayensis was found prevalent in numerous urban parks of Singapore [14]. Not only were Ae. malayensis eggs collected in the forested areas of the parks but they were also found in some open-air areas frequented by humans [14]. The current study reports that in 2018, the National Vector Surveillance program in Singapore detected the presence of Ae. malayensis in Gravitraps [21] outside the urban parks of the island. Although the majority of *Aedes* mosquitoes found in public apartment blocks were Ae. aegypti, Ae. malayensis was detected in Gravitraps deployed in high-rise building areas, underlining the potential of this mosquito species to colonize urban habitats. It was recently hypothesized that peridomestic mosquito species, such as Ae. malayensis, which escape vector control measures targeting domestic vector species, may have contributed to the (re-)emergence of arboviruses in Singapore in the last two decades [14]. The well-known peridomestic mosquito Ae. albopictus, which was highly abundant in our human-baited trap survey, significantly increases the risk of arbovirus transmission in Singapore [14]. Control of peridomestic mosquitoes is a challenge because in addition to artificial breeding sites, less accessible natural sites such as tree holes need to be targeted as well [31].

Although we did not measure other parameters that influence the vectorial capacity of a mosquito species, such as adult density and survival, and the duration of the virus extrinsic incubation period [32], vector competence and attraction to humans make *Ae. malayensis* a potential vector of YFV and other arboviruses that deserves increased attention from public health authorities. The level of *Ae. malayensis* vector competence for YFV that we observed was in the same range as our *Ae. aegypti* controls from Laos, and as Brazilian *Ae. aegypti* that were orally challenged with the same virus isolate in a previous study [33]. It is worth noting, however, that vector competence results cannot be readily compared if they did not use the same oral infectious dose, or if other factors such as virus strain and detection assays were different. The dose-response curves that we generated in the present study will be useful for

future comparisons because they provide absolute estimates of vector competence indices, such as the OID₅₀. Moreover, two vector species or populations may have a similar level of vector competence at a given infectious dose, but vary at a different dose. For example, the DENV type 2 susceptibility of two field-derived populations of *Ae. aegypti* in Thailand were similar at a medium dose, but different at a low dose [34]. This example indicates that the minimal threshold for infection may differ between mosquito populations. Likewise, in the present study, we observed similar YFV OID₅₀ values for *Ae. aegypti* and *Ae. malayensis* but *Ae. aegypti* had a lower YFV OID₁₀ value. In natural conditions, the oral infectious dose is the primary driver of successful human-to-mosquito transmission of arboviruses [35,36].

Despite an efficient vaccine, YFV still remains an important public health concern in Africa and South America. YFV is maintained in sylvatic cycles between non-human primates and arboreal mosquitoes, and human outbreaks occur when the virus is introduced into an urban cycle mediated by Ae. aegypti [37]. Although there is no evidence of a sylvatic cycle of YFV in the Asia-Pacific region, the risk of spillover transmission exists. The lack of YFV outbreaks in the Asia-Pacific region to date could be due to cross-protective immunity conferred by other widespread flaviviruses such as DENV and Japanese encephalitis virus, but it could also reflect relatively fewer opportunities for YFV introduction until now. Rampant globalization now puts the Asia-Pacific region at an unprecedented risk of YFV introduction [6-8]. Billions of immunologically naïve people, together with competent Ae. aegypti populations [38,39], create a dangerous set of conditions that are theoretically favorable to a massive YFV epidemic. On the other hand, a scenario whereby YFV would establish a novel sylvatic cycle [40] seems unlikely in Singapore given the small size of resident non-human primate populations. Clearly, places with high densities of the domestic vector Ae. aegypti are most at risk of spillover transmission, but we emphasize that in the absence of Ae. aegypti, ancillary vectors should not be overlooked. Competent mosquito species such as Ae. malayensis could facilitate introduction of YFV into the Asia-Pacific region. In Singapore, strict and sustained vector control measures have achieved low densities of Ae. aegypti [9] but this has failed to prevent arbovirus outbreaks in the last two decades [10]. Following a previous study on the ability of *Ae. malayensis* from Singapore to transmit DENV and CHIKV [14], we reiterate the need to account for peridomestic species in the management of mosquito vector populations.

Acknowledgments

We thank Catherine Lallemand for assistance with mosquito rearing and three anonymous reviewers for constructive comments on an earlier version of the manuscript. We are grateful to the blood donor volunteers for participation in this study and the ICAReB staff for its support. We thank the National Park Board Singapore for their permission to collect in their parks and Jeffrey C. Hertz for his support and advice with the mosquito collections.

Author Contributions

Conceptualization: Elliott F. Miot, Paul T. Brey, Louis Lambrechts.

Formal analysis: Elliott F. Miot, Louis Lambrechts.

Funding acquisition: Paul T. Brey, Louis Lambrechts.

Investigation: Elliott F. Miot, Fabien Aubry, Stéphanie Dabo, Cheong H. Tan, Lee C. Ng.

Resources: Ian H. Mendenhall, Sébastien Marcombe, Anna-Bella Failloux, Julien Pompon, Louis Lambrechts.

Supervision: Paul T. Brey, Louis Lambrechts.

Visualization: Elliott F. Miot.

Writing - original draft: Elliott F. Miot, Louis Lambrechts.

Writing – review & editing: Elliott F. Miot, Anna-Bella Failloux, Julien Pompon, Paul T. Brey, Louis Lambrechts.

References

- 1. World Health Organization (WHO). WHO | WHO Report on Global Surveillance of Epidemic-prone Infectious Diseases—Yellow fever. WHO. World Health Organization. Available: https://www.who.int/ csr/resources/publications/yellowfev/CSR_ISR_2000_1/en/
- 2. World Health Organization (WHO). Yellow Fever—Fact sheets. In: WHO [Internet]. 1 May 2018 [cited 29 Apr 2019]. Available: https://www.who.int/en/news-room/fact-sheets/detail/yellow-fever
- Grobbelaar AA, Weyer J, Moolla N, Jansen van Vuren P, Moises F, Paweska JT. Resurgence of Yellow Fever in Angola, 2015–2016. Emerging Infect Dis. 2016; 22: 1854–1855. <u>https://doi.org/10.3201/</u> eid2210.160818 PMID: 27536787
- World Health Organization (WHO). Emergencies Preparedness, Response: Yellow Fever–Nigeria [Internet]. 2019 [cited 29 Apr 2019]. Available: https://www.who.int/csr/don/09-january-2019-yellow-fever-nigeria/en/
- WHO | Yellow fever–Brazil. WHO. World Health Organization; 2019. Available: https://www.who.int/ csr/don/18-april-2019-yellow-fever-brazil/en/
- Gubler DJ. Pandemic yellow fever: a potential threat to global health via travelers. J Travel Med. 2018; 25: 160. https://doi.org/10.1093/jtm/tay097 PMID: 30299494
- World Health Organization (WHO). Emergencies preparedness, response: Yellow Fever–China. In: WHO [Internet]. World Health Organization; 22 Apr 2016 [cited 3 Apr 2019]. Available: <u>http://www.who.</u> int/csr/don/22-april-2016-yellow-fever-china/en/
- Butler D. Fears rise over yellow fever's next move. Nature. 2016.: 155–156. <u>https://doi.org/10.1038/532155a PMID: 27075072</u>
- Egger JR, Ooi EE, Kelly DW, Woolhouse ME, Davies CR, Coleman PG. Reconstructing historical changes in the force of infection of dengue fever in Singapore: implications for surveillance and control. Bull World Health Organ. World Health Organization; 2008; 86: 187–196. https://doi.org/10.2471/BLT. 07.040170 PMID: 18368205
- Hapuarachchi HC, Koo C, Rajarethinam J, Chong C-S, Lin C, Yap G, et al. Epidemic resurgence of dengue fever in Singapore in 2013–2014: A virological and entomological perspective. BMC Infect Dis. BioMed Central; 2016; 16: 300. https://doi.org/10.1186/s12879-016-1606-z PMID: 27316694
- 11. Koh BKW, Ng L-C, Kita Y, Tang CS, Ang LW, Wong KY, et al. The 2005 dengue epidemic in Singapore: epidemiology, prevention and control. Ann Acad Med Singap. 2008; 37: 538–545. PMID: 18695764
- 12. Leo YS, Chow ALP, Tan LK, Lye DC, Lin L, Ng LC. Chikungunya outbreak, Singapore, 2008. Emerging Infect Dis. 2009; 15: 836–837. https://doi.org/10.3201/eid1505.081390 PMID: 19402989
- Maurer-Stroh S, Mak T-M, Ng Y-K, Phuah S-P, Huber RG, Marzinek JK, et al. South-east Asian Zika virus strain linked to cluster of cases in Singapore, August 2016. Euro Surveill. European Centre for Disease Prevention and Control; 2016; 21: 1232. https://doi.org/10.2807/1560-7917.ES.2016.21.38.30347 PMID: 27684526
- Mendenhall IH, Manuel M, Moorthy M, Lee TTM, Low DHW, Missé D, et al. Peridomestic Aedes malayensis and Aedes albopictus are capable vectors of arboviruses in cities. PLoS Negl Trop Dis. 2017; 11: e0005667. https://doi.org/10.1371/journal.pntd.0005667 PMID: 28650959
- Rodhain F, Hannoun C, Jousset FX, Ravisse P. Isolation of the yellow fever virus in Paris from 2 imported human cases. Bull Soc Pathol Exot Filiales. 1979; 72: 411–415. PMID: 261928
- Fontaine A, Jiolle D, Moltini-Conclois I, Lequime S, Lambrechts L. Excretion of dengue virus RNA by Aedes aegypti allows non-destructive monitoring of viral dissemination in individual mosquitoes. Sci Rep. 2016; 6: 24885. https://doi.org/10.1038/srep24885 PMID: 27117953
- Galun R, Avi-Dor Y, Bar-Zeev M. Feeding Response in Aedes aegypti: Stimulation by Adenosine Triphosphate. Science. 1963; 142: 1674–1675. https://doi.org/10.1126/science.142.3600.1674 PMID: 17834375
- Aitken THG. An in vitro feeding technique for artificially demonstrating virus transmission by mosquitoses. Mosquito News. 1977; 37: 130–133.

- Petrova E, Gracias S, Beauclair G, Tangy F, Jouvenet N. Uncovering Flavivirus Host Dependency Factors through a Genome-Wide Gain-of-Function Screen. Viruses. Multidisciplinary Digital Publishing Institute; 2019; 11: 68. https://doi.org/10.3390/v11010068 PMID: 30650657
- 20. Tangena J-AA, Thammavong P, Hiscox A, Lindsay SW, Brey PT. The Human-Baited Double Net Trap: An Alternative to Human Landing Catches for Collecting Outdoor Biting Mosquitoes in Lao PDR. PLoS ONE. 2015; 10: e0138735. https://doi.org/10.1371/journal.pone.0138735 PMID: 26381896
- Lee C, Vythilingam I, Chong C-S, Abdul Razak MA, Tan C-H, Liew C, et al. Gravitraps for management of dengue clusters in Singapore. Am J Trop Med Hyg. 2013; 88: 888–892. <u>https://doi.org/10.4269/ aitmh.12-0329 PMID: 23478581</u>
- 22. Fox J, Weisberg S. An R Companion to Applied Regression. SAGE Publications; 2011.
- 23. Venables WN, Ripley BD. Modern Applied Statistics with S. Springer Science & Business Media; 2003.
- 24. R Core Team. R: A Language and Environment for Statistical Computing [Internet]. Vienna, Austria; 2018. Available: https://www.R-project.org/
- 25. Wickham H. ggplot2. Cham: Springer; 2016. https://doi.org/10.1007/978-3-319-24277-4
- Kassambara A. ggpubr: "ggplot2" Based Publication Ready Plots [Internet]. 2018. pp. GPL–2. Available: http://www.sthda.com/english/rpkgs/ggpubr
- Huang Y-M. Contributions to the Mosquito Fauna of Southeast Asia. XIV. The Subgenus Stegomyia of Aedes in Southeast Asia I—The Scutellaris Group of Species. Contr Am Entomol Inst. 1972; 1–109.
- Tewari SC, Hiriyan J, Reuben R. Epidemiology of subperiodic Wuchereria bancrofti infection in the Nicobar Islands, India. Trans R Soc Trop Med Hyg. 1995; 89: 163–166. https://doi.org/10.1016/0035-9203 (95)90478-6 PMID: 7778141
- 29. Motoki MT, Miot EF, Rueda LM, Vongphayloth K, Phommavanh N, Lakeomany K, et al. First Record of Aedes (Stegomyia) malayensis Colless (Diptera: Culicidae) in the Lao People's Democratic Republic, Based on Morphological Diagnosis and Molecular Analysis. US Army Med Dep J. 2018;: 1–7.
- Rattanarithikul R, Harbach RE, Harrison BA, Panthusiri P, Coleman RE, Richardson JH. Illustrated keys to the mosquitoes of Thailand. VI. Tribe Aedini. Southeast Asian J Trop Med Public Health. 2010; 41 Suppl 1: 1–225.
- Itoh T, Kawada H, Abe A, Eshita Y, Rongsriyam Y, Igarashi A. Utilization of bloodfed females of Aedes aegypti as a vehicle for the transfer of the insect growth regulator pyriproxyfen to larval habitats. J Am Mosg Control Assoc. 1994; 10: 344–347. PMID: 7807075
- Kramer LD, Ebel GD. Dynamics of flavivirus infection in mosquitoes. Adv Virus Res. 2003; 60: 187– 232. PMID: 14689695
- Couto-Lima D, Madec Y, Bersot MI, Campos SS, Motta M de A, Santos FBD, et al. Potential risk of reemergence of urban transmission of Yellow Fever virus in Brazil facilitated by competent Aedes populations. Sci Rep. 2017; 7: 4848. https://doi.org/10.1038/s41598-017-05186-3 PMID: 28687779
- Pongsiri A, Ponlawat A, Thaisomboonsuk B, Jarman RG, Scott TW, Lambrechts L. Differential Susceptibility of Two Field Aedes aegypti Populations to a Low Infectious Dose of Dengue Virus. PLoS ONE. 2014; 9: e92971. https://doi.org/10.1371/journal.pone.0092971 PMID: 24664142
- Duong V, Lambrechts L, Paul RE, Ly S, Lay RS, Long KC, et al. Asymptomatic humans transmit dengue virus to mosquitoes. Proc Natl Acad Sci USA. 2015; 112: 14688–14693. <u>https://doi.org/10.1073/</u> pnas.1508114112 PMID: 26553981
- Nguyet MN, Duong THK, Trung VT, Nguyen THQ, Tran CNB, Long VT, et al. Host and viral features of human dengue cases shape the population of infected and infectious Aedes aegypti mosquitoes. Proc Natl Acad Sci USA. 2013; 110: 9072–9077. https://doi.org/10.1073/pnas.1303395110 PMID: 23674683
- Barrett AD, Monath TP. Epidemiology and ecology of yellow fever virus. Adv Virus Res. 2003; 61: 291– 315. PMID: 14714435
- Aitken TH, Downs WG, Shope RE. Aedes aegypti strain fitness for yellow fever virus transmission. Am J Trop Med Hyg. 1977; 26: 985–989. https://doi.org/10.4269/ajtmh.1977.26.985 PMID: 907058
- Tabachnick WJ, Wallis GP, Aitken TH, Miller BR, Amato GD, Lorenz L, et al. Oral infection of Aedes aegypti with yellow fever virus: geographic variation and genetic considerations. Am J Trop Med Hyg. 1985; 34: 1219–1224. https://doi.org/10.4269/ajtmh.1985.34.1219 PMID: 3834804
- 40. Hanley KA, Monath TP, Weaver SC, Rossi SL, Richman RL, Vasilakis N. Fever versus fever: the role of host and vector susceptibility and interspecific competition in shaping the current and future distributions of the sylvatic cycles of dengue virus and yellow fever virus. Infect Genet Evol. 2013; 19: 292–311. https://doi.org/10.1016/j.meegid.2013.03.008 PMID: 23523817